Author’s response to reviews

Title: Trajectories of physical function prior to death and brain neuropathology in a community-based cohort: the ACT study

Authors:
Andrea LaCroix (alacroix@ucsd.edu)
Rebecca Hubbard (hubbard.r@ghc.org)
Shelly Gray (slgray@uw.edu)
Melissa Anderson (anderson.melissa@ghc.org)
Paul Crane (pcrane@u.washington.edu)
Joshua Sonnen (joshua.sonnen@utah.edu)
Oleg Zaslavsky (ozasl@uw.edu)
Eric Larson (larson.e@ghc.org)

Version: 2 Date: 31 Aug 2017

Author’s response to reviews:
August 28, 2017

BMC Geriatrics Section Editors
Neurology, stroke and cognition
Pieter Jelle Visser, Vrije Universiteit Medical Center, Netherlands
Physical functioning, physical health and activity
Ian Cameron, University of Sydney, Australia
Dear BMC Geriatrics Editorial Office:

In response to your communication on August 2, 2017, we have prepared a revised version of our manuscript entitled, “Trajectories of physical function prior to death and brain neuropathology in a community-based cohort: the ACT study”. Specifically, we added a more complete description of the neuropathology methods in the Outcomes section of manuscript as requested by Associate Editor Hiroyuki Shimada (lines 155-178).

Once again, we are grateful for your consideration of this revised manuscript and look forward to the results of your editorial decision.

Sincerely,

Andrea Z. LaCroix, PhD
Professor and Chief of Epidemiology
Director, Women’s Health Center of Excellence
Department of Family Medicine and Public Health
University of California, San Diego